Author’s response to reviews

Title: Establishing the added benefit of measuring MMP9 in FOB positive patients as a part of the Wolverhampton colorectal cancer screening programme

Authors:

  Sue Wilson (s.wilson@bham.ac.uk)
  Taina Taskila (T.K.Taskila@bham.ac.uk)
  Tariq Ismail (tariq.ismail@uhb.nhs.uk)
  Deborah D Stocken (d.d.stocken@bham.ac.uk)
  Ashley Martin (a.martin@bham.ac.uk)
  Val Redman (v.d.redman@bham.ac.uk)
  Michael Wakelam (michael.wakelam@bbsrc.ac.uk)
  Ian Perry (perryhome@blueyonder.co.uk)
  Richard Hobbs (f.d.r.hobbs@bham.ac.uk)

Version: 2 Date: 21 January 2009

Author’s response to reviews:

In the Authors’ contribution section (p.5); we have added Dr. Richard Hobb’s initials into the first phrase: "SW, TI, MW, RH and IP designed the manuscript."